Author's response to reviews

Title: Gastric and duodenal metastases of malignant melanoma: a case report

Authors:

- Vedat Goral Prof (vegoral@hotmail.com)
- Feyzullah Ucmak PhD (ucmakfeyz@hotmail.com)
- Sezgin Barutcu Dr (sezginbarutcu@yahoo.com)
- Serdar Yildirim Dr (yildirs@hotmail.com)
- Serdar Ileri Dr (serdari@dicle.edu.tr)
- Huseyin Buyukbayram Prof (husram@dicle.edu.tr)

Version: 8 Date: 25 September 2010

Author's response to reviews: see over
ANSWERS TO THE REVIEWER’S REPORT

Dear Carsten Nieder

1) The reference 5 was noted in the text.
2) The reference 12 was changed with another literature.
3) Abbreviations were explained in the text.
4) I used female patient term in the text.
5) All malign melanoma terms were changed as ‘malignant melanoma’ in the text.
6) The patient had not B-symptoms.
7) The details of patient was added in the text.
8) The patient was referred to our oncology clinic after the diagnosis for chemotherapy.
9) Therefore, every metastatic malign melanoma cases should be... term were changed as, Therefore, metastatic malignant melanoma cases should be...

Thank you very much for revisions. It was very important and helpful for mine article.
Sincerely yours.

Prof. Dr. Vedat Goral
Dicle University School of Medicine
Department of Gastroenterology
Diyarbakir/Turkey
Author's response to reviewers (2)
Title: GASTRIC AND DUODENAL METASTASES OF MALIGNANT MELANOMA

Authors: Vedat Goral (1), Feyzullah Ucmak (1), Serdar Yildirim (2), Sezgin Barutcu (2), Serdar Ileri (2), Huseyin Buyukbayram (3)
Dicle University School of Medicine Department of Gastroenterology (1), Internal Medicine (2) and Pathology (3). Diyarbakir/Turkey
vegoral@hotmail.com.

Version: 2 Date: 08.June.2010
Author's response to reviews: see over

Dear Marek Wronski,

1) The primary focus of this case was left thumb. She had amputation history from left thumb six years ago. This was noted in the article.
2) It is true. She was at the late-stage disease. GI metastases usually occurs at the late stage like our case report.
3) Porta hepatis, metastatic lymphadenopathy (LAP), and thorax CT terms were corrected. A few circles in the…. sentences were canceled.
4) All malign melanoma terms were changed as ‘malignant melanoma’ in the article.
5) The references format had changed.
6) The reference 12 was changed. I put another new literature.
7) The technical parameters of CT were omitted.
8) The abbreviations were changed with explanation in the article.

Thank you very much for revisions. It was very important for mine article.
Sincerely yours.

Prof. Dr. Vedat Goral
Dicle University School of Medicine
Department of Gastroenterology
Diyarbakir/Turkey